Author's response to reviews

Title: Effectiveness of guideline-based care by occupational physicians on the return-to-work of workers with common mental disorders: design of a cluster-randomised controlled trial.

Authors:

Karlijn M. van Beurden (k.vanbeurden@tilburguniversity.edu)
Evelien P.M. Brouwers (e.p.m.brouwers@tilburguniversity.edu)
Margot C.W. Joosen (m.c.w.joosen@tilburguniversity.edu)
Berend Terluin (b.terluin@vumc.nl)
Jac J.L. van der Klink (j.j.l.van.der.klink@umcg.nl)
Jaap van Weeghel (j.vanweeghel@tilburguniversity.edu)

Version: 2 Date: 28 January 2013

Author's response to reviews:

Editorial Board

BioMed Central Public Health

Natalie Pafitis, Executive Editor

Tilburg, 28th January, 2013

Dear Ms Pafitis, Dr Dizon,

Please find enclosed our manuscript entitled ‘Effectiveness of guideline-based care by occupational physicians on the return-to-work of workers with common mental disorders: design of a cluster-randomised controlled trial.’ and the additional file ‘Dutch questionnaire for employers’.

We refer to the e-mail exchange between Evelien Brouwers and Natalie Pafitis (28 January 2013) about the requested changes to the formatting of the manuscript requested by Jimmar Dizon. Natalie Pafitis wrote: ‘Please accept my apologies for any confusion caused in our request regarding the questionnaires to be administered in your study. We ask that you please submit, as additional files, any copies of questionnaires which have been designed for the purpose of your study. Please note that these do not have to be translated in English.’ Therefore, we upload to the additional files the ‘Dutch questionnaire for employers’.

We hope this provides the requested changes and we hope you will consider this manuscript for publication in your journal.
On behalf of all authors, yours sincerely,

Karlijn M. van Beurden, MSc
Tilburg University, department Tranzo, Scientific center for care and welfare
PO Box 90153
5000 LE Tilburg
The Netherlands